Author's response to reviews

Title: Self-perceived competence correlates poorly with objectively measured competence in Evidence Based Medicine among medical students

Authors:

Nai Ming Lai (lainm123@yahoo.co.uk)
Cheong Lieng Teng (tengcl@gmail.com)

Version: 3 Date: 27 May 2011

Author's response to reviews:

26 May 2011

Professor Ian Wilson,
Associate Editor,
BMC Medical Education.

Dear Prof Wilson,

MS: 2059356211503123

Self-perceived competence correlates poorly with objectively measured competence in Evidence Based Medicine among medical students

Nai Ming Lai and Cheong Lieng Teng

We are pleased to inform you that we have completed the following formatting changes in the final proof of our manuscript:

1. Submitting identical abstracts in the submission system and the manuscript.
2. Changing the heading in the main text from "Introduction" to “Background”.
3. Adding a statement that both authors read and approved the final manuscript in Authors’ Contributions.
4. Removing all the formatting marks from the Track Changes tool.
5. Checking spelling and grammar throughout the text and revising accordingly.

Sincerely,

____________________
Nai Ming Lai
Lead author